Author's response to reviews

Title: Implementing hospital guidelines improves warfarin use in non-valvular atrial fibrillation: a before-after study.

Authors:

Simona Bo (sbo@molinette.piemonte.it)
Susanna Valpreda (valpinax@hotmail.com)
Luca Scaglione (lscaglio@tin.it)
Daniela Boscolo (sbo@molinette.piemonte.it)
Marina Piobbici (sbo@molinette.piemonte.it)
Mario Bo (mario.bo@unito.it)
Giovannino Ciccone (gianni.ciccone@cpo.it)

Version: 5 Date: 18 July 2007

Author's response to reviews:

Dear Editor,

the following changes have been made, according to your request:

The country in the affiliation details has been included.
Websites have been removed from the text.
All red text has been changed to black.
The underlining from the last sentence of the Authors' contribution paragraph has been removed.